Author’s response to reviews

Title: Does the world need a scientific society for research on how to improve healthcare?

Authors:

Michel Wensing (M.Wensing@iq.umcn.nl)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Martin P Eccles (martin.eccles@newcastle.ac.uk)

Version: 3 Date: 2 February 2012

Author’s response to reviews:

All suggested changes have been accepted. Three items need commenting:
-the name of the Nijmegen organisation is "institute" not "centre"
-I suggest to leave out headings in the abstract, because most is Discussion
-BMJ Quality and Safety - I propose to stick to BMJ as this is what they use

Michel